Author’s response to reviews

Title: Identification and characterization of a novel 43-bp deletion mutation of the ATP7B gene in a Chinese patient with Wilson's disease: a case report

Authors:
Zhengfeng Xu (zhengfeng_xu_nj@163.com)
Gang Liu (liugang_516@126.com)
Dingyuan Ma (dingyuanma2002@163.com)
Jian Cheng (13770571912@163.com)
Jingjing Zhang (18913384682@163.com)
Chunyu Luo (lcy0929@126.com)
Yun Sun (sunyun80@126.com)
Ping Hu (njfybyhuping@163.com)
Yuguo Wang (wyg.num1@163.com)
Tao Jiang (jiangzhang784@163.com)

Version: 2 Date: 12 Mar 2018

Author’s response to reviews:

Response to Editor’ Comments

Editor Comment 1: Please represent authors' names using their initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Response: We corrected it in the revised manuscript. Thank you. (line 238–242, Page 12)

Editor Comment 2: We would also like to ask for you to provide more justification for the contributions of Yuguo Wang (YW), Yun Sun (YS), Ping Hu (PH), as currently they do not automatically qualify for authorship. Contribution to one aspect of the manuscript, alone, does not usually justify authorship.
Response: These people had specific contributions for this study, so they should be classified as authors. And, a more detailed description of their work is given in the revised manuscript. (line 239-241, Page 12)

Editor Comment 3: Please clarify in the manuscript whether consent was written or verbal.

Response: It was clear that written informed consent was obtained and had been stated in the revised manuscript. (line 246, Page 12)

Editor Comment 4: Please remove any additional files which you do not want published as they are no longer needed at this stage, including the CARE checklist. Please ensure that you include a statement in the Case presentation section clarifying that the CARE guidelines were followed in reporting this case.

Response: We corrected it in the revised manuscript. Thank you. (line 118, Page 6)

Editor Comment 5: At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We do upload our manuscript as a single, final, clean version and ensure that all relevant tables/figures/additional files also are clean versions.